Author's response to reviews

Title: Accuracy of Parent-Reported Information for Estimating Prevalence of Overweight and Obesity in a Race-Ethnically Diverse Pediatric Clinic Population Aged 3 to 12

Authors:

   Nancy P Gordon (nancy.gordon@kp.org)
   R. Grant Mellor (r.grant.mellor@kp.org)

Version: 6  Date: 21 January 2015

Author's response to reviews: see over
Dear Editor:

We are re-submitting a revised version of our manuscript titled “Accuracy of Parent-Reported Information for Estimating Prevalence of Overweight and Obesity in a Race-Ethnically Diverse Pediatric Clinic Population Aged 3 to 12”.

Per your request, we have:

- Removed the figure number and title from the figure and put the figure legend in the text following the References as requested.
- Removed the survey questionnaire from the text, and will submit it as supplemental material instead.

We hope that with these revisions you will now find the manuscript acceptable for publication.

Sincerely,

Nancy P. Gordon, ScD
Research Scientist

Kaiser Permanente Division of Research
2000 Broadway, Oakland, CA 94612
ph: (510) 891-3587
fax: (510) 891-3606
email: Nancy.Gordon@kp.org